Author's response to reviews

Title: Blood donors’ motivation and attitudes to non-remunerated blood donorship in Lithuania

Authors:

Ilona Buciuniene (ilona.buciuniene@ism.lt)
Laimute Stoniene (l.stoniene@kraujodonoryste.lt)
Aurelija Blazeviciene (blazeviciene@yahoo.com)
Ruta Kazlauskaite (ruta.kazlauskaite@ism.lt)
Vida Skudiene (vida.skudiene@ism.lt)

Version: 4  Date: 14 June 2006

Author’s response to reviews:

Dear Iratxe Puebla, Senior Assistant Editor
BMC-series journals,

We would like to thank you for taking your time to review our article. We have revised the article and change Authors' contributions and Acknowledgements parts. Also we revised all article and correct small technical mistake. Finally, we would like to inform, that we have ethical approval for the study from Bioethics committee. Respectfully,

Authors